Author's response to reviews

Title: U.S. medical resident familiarity with national tuberculosis guidelines

Authors:

Petros C Karakousis (petros@jhmi.edu)
Frangiscos G Sifakis (fsifakis@jhsph.edu)
Ruben Montes de Oca (montesd@jhsph.edu)
Valerianna C Amorosa (amorosav@comcast.net)
Kathleen R Page (kpage2@jhmi.edu)
Yuykari C Manabe (ymanabe@jhmi.edu)
James D Campbell (Jcampbel@medicine.umaryland.edu)

Version: 3 Date: 18 May 2007

Author's response to reviews: see over
Dear Dr. Kouremenou,

Please find attached the revised version of the article entitled “U.S. medical resident familiarity with national tuberculosis guidelines”, which we wish to submit for publication in *BMC Infectious Diseases*.

We have addressed each of the referees’ comments in the attached point-by-point responses.

This manuscript represents original research that has not been published and is not under consideration elsewhere. All of the authors participated in the preparation of the manuscript and have agreed to its publication in the *BMC Infectious Diseases*.

Thank you very much for your consideration.

Sincerely yours,

Petros C. Karakousis, M.D.
Assistant Professor